Author's response to reviews

Title: Differential misclassification of confounders in comparative evaluation of hospital care quality: caesarean sections in Italy.

Authors:

Mirko Di Martino (m.dimartino@deplazio.it)
Danilo Fusco (d.fusco@deplazio.it)
Paola Colais (p.colais@deplazio.it)
Luigi Pinnarelli (l.pinnarelli@deplazio.it)
Marina Davoli (m.davoli@deplazio.it)
Carlo A Perucci (perucci@agenas.it)

Version: 3 Date: 26 November 2013

Author's response to reviews:

In the new submission, the following sections have been modified and corrected, according to the Journal Editorial Office suggestion: title page, acknowledgements, figures and references. We also changed the sequence of the sections of the article.

Looking forward to receiving news,

Best regards,

Mirko Di Martino

Please address all correspondence to:
Mirko Di Martino
Department of Epidemiology, Lazio Regional Health Service.
Via Santa Costanza, 53
00198 Roma, Italy.
Phone: --39-06 83060364
Fax: --39-06 83060374
E-mail: m.dimartino@deplazio.it.